Author's response to reviews

Title: Evaluation of Tocilizumab Therapy in Patients with Rheumatoid Arthritis Based on FDG-PET/CT

Authors:

Koichi Okamura (kokamura@gunma-u.ac.jp)
Yukio Yonemoto (yukioyone@yahoo.co.jp)
Chisa Okura (chibutaman@gmail.com)
Tetsuya Higuchi (tetsuyah92md@gmail.com)
Yoshito Tsushima (tsushima.rad.gunma@gmail.com)
Kenji Takagishi (ktakagishi@gunma-u.ac.jp)

Version: 4 Date: 7 November 2014

Author's response to reviews: see over
1. Please reformat the title page at the front of your manuscript file. It should contain, at minimum, the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting author.

I added email address of all authors.

2. Name of ethics committee:

Please update your ethics statement to include the name of the ethics committee that approved your study.

Institutional Review Board of Gunma University Hospital

3. By way of a section ‘Acknowledgements’, please acknowledge anyone who contributed towards the article by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship.

We added “Acknowledgements” section in the manuscripts.

Please also include the source(s) of funding for each author, and for the manuscript preparation. Authors must describe the role of the funding body, if any, in design, in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication. Please also acknowledge anyone who contributed materials essential for the study.

We received no funding support to this study. This study was performed by our University’s budget.

If a language editor has made significant revision of the manuscript, we recommend that you acknowledge the editor by name, where possible.

The language editor (native English check) did not made significant revision of the
The role of a scientific (medical) writer must be included in the acknowledgements section, including their source(s) of funding. We suggest wording such as 'We thank Jane Doe who provided medical writing services on behalf of XYZ Pharmaceuticals Ltd.' Authors should obtain permission to acknowledge from all those mentioned in the Acknowledgements section.

We ordered English proofreading service (native English check) before submitting, however we did not use medical writer. We wrote our manuscripts by ourselves.

4. In addition, could you please clarify if the patients received the Tocilizumab therapy as part of their standard care. If so, could you please state this into the method section of your manuscript.

We added “and TCZ therapy was performed as part of their standard care.” in the method section.